Author's response to reviews

Title: Acute mercury poisoning presenting as fever of unknown origin in an adult woman: a case report

Authors:

Gonul Cicek-Senturk (drgonulcicek@hotmail.com)
Fatma Aybala Altay (aybalaaltay@hotmail.com)
Aysegul Ulu-Kilic (draysegululu@yahoo.com)
Yunus Gurbuz (gurbuyunus@hotmail.com)
Ediz Tutuncu (ediztutuncu@gmail.com)
Irfan Sencan (isencanibu@yahoo.com)

Version: 4 Date: 16 March 2014

Author's response to reviews:

Dear editor,
Thank you for consideration and patience of our manuscript for publication in your journal.
I apologize for my carelessness.
We put the title page, a list of abbreviations (after the Conclusion section) and an acknowledgement section (before the reference list) in the manuscript.
We removed the figure legends and cropped the figures as you wanted
We put the Figure after the References list.
We hope all the changes have been made as you listed.
Our best wishes,
Authors:
Gonul Cicek-Senturk
Fatma Aybala Altay
Aysegul Ulu-Kilic
Yunus Gurbuz
Ediz Tutuncu
Irfan Sencan